\documentclass[reprint,amsmath,amssymb]{revtex4-1}

\usepackage{color,epsfig,rotating,graphicx}
\usepackage{amsmath,amssymb,amscd}


\usepackage[latin1]{inputenc}
\usepackage[english]{babel}

\usepackage{dsfont}
\usepackage{textcomp}
\usepackage{setspace}

% Eigene Befehle
\renewcommand{\Re}{{\rm Re}}
\renewcommand{\Im}{{\rm Im}}

\newcommand{\ri}{{\rm i}}
\newcommand{\re}{{\rm e}}
\newcommand{\rd}{{\rm d}}

\newcommand{\rp}{{\rm p}}
\newcommand{\rs}{{\rm s}}
\newcommand{\rr}{{\rm r}}


\newcommand{\kb}{k_{\rm B}}
\newcommand{\Tr}{{\rm Tr}}


\begin{document}
\begin{thebibliography}{99}
  %Heat flux
  \bibitem{PvH1971} D. Polder and M. van Hove, Phys. Rev. B {\bf 4}, 3303 (1971).
	%Near field thermal radiation review
	\bibitem{ZhangReview2009} S. Basu, Z. M. Zhang, and C. J. Fu, Int. J. Energy Res. {\bf 33}, 1203 (2009).
	\bibitem{ZhangReview2013} K. Park, and Z. Zhang,  Front. Heat Mass Transfer {\bf 4}, 013001 (2013).
  %experiment
  \bibitem{HuEtAl2008} L. Hu, A. Narayanaswamy, X. Chen, and G. Chen, Appl. Phys. Lett. {\bf 92}, 133106 (2008).
  \bibitem{ShenEtAl2009} S. Shen, A. Narayanaswamy, and G. Chen, Nano Lett. {\bf 9}, 2909 (2009).
  \bibitem{Kralik2012} T. Kralik, P. Hanzelka, M. Zobac, V. Musilova, T. Fort, and M. Horak, Phys. Rev. Lett. {\bf 109}, 224302 (2012)
  %surface modes
  \bibitem{JoulainEtAl2005} K. Joulain, J.-P. Mulet, F. Marquier, R. Carminati, and J.-J. Greffet, Surf. Sci. Rep. {\bf 57}, 59 (2005).
  \bibitem{BiehsEtAl2010} S.-A. Biehs, E. Rousseau, and J.-J. Greffet, Phys. Rev. Lett. {\bf 105}, 234301 (2010). 
  %HF hyperbolic modes
  \bibitem{Nefedov2011} I. S. Nefedov and C. R. Simovski, Phys. Rev. B {\bf 84}, 195459 (2011).
  \bibitem{BiehsEtAl2012} S.-A. Biehs, M. Tschikin, and P. Ben-Abdallah, Phys. Rev. Lett. {\bf 109}, 104301 (2012).
  \bibitem{GuoEtAl2012} Y. Guo, C. L. Cortes, S. Molesky, and Z. Jacob, Appl. Phys. Lett. {\bf 101}, 131106 (2012).
  %touchless cooling
  \bibitem{Ottens2011} R. S. Ottens, V. Quetschke, S. Wise, A. A. Alemi, R. Lundock, G. Mueller, D. H. Reitze, D. B. Tanner,and B. F. Whiting, Phys. Rev. Lett. {\bf 107}, 014301 (2011).
  \bibitem{GuhaEtAl2012} B. Guha, C. Otey, C.B. Poitras, S. Fan, and M. Lipson, Nano Lett. {\bf 12}, 4546 (2012).
  %thermal imaging
\bibitem{Yannick} Y. De Wilde, F. Formanek, R. Carminati, B. Gralak, P.A. Lemoine, K. Joulain, J.P. Mulet, Y. Chen, and J.J. Greffet,  Nature {\bf 444}, 740 (2006).
  \bibitem{KittelEtAl2008} A. Kittel , U. Wischnath , J. Welker , O. Huth , F. R\"{u}ting, and S.-A. Biehs, Appl. Phys. Lett. {\bf 93}, 193109 (2008).
  \bibitem{HuthEtAl2011} F. Huth, M. Schnell, J. Wittborn, N. Ocelic and R. Hillenbrand, Nat. Mater. {\bf 10}, 352 (2011).
  \bibitem{WorbesEtAl2013} Ludwig Worbes, David Hellmann, and Achim Kittel, Phys. Rev. Lett. {\bf 110}, 134302 (2013).
  %TPV
  \bibitem{MatteoEtAl2001} R. S. DiMatteo, P. Greiff, S. L. Finberg, K. A. Young-Waithe, H. K. Choy, M. M. Masaki, and C. G. Fonstad, Appl. Phys. Lett. \textbf{79}, 1894 (2001).
  \bibitem{NarayanaswamyChen2003} A. Narayanaswamy and G. Chen, Appl. Phys. Lett. \textbf{82}, 3544 (2003).
  \bibitem{LarocheEtAl06} M. Laroche, R. Carminati, and J.-J. Greffet, J. Appl. Phys. {\bf 100}, 063704 (2006).
  \bibitem{ParkEtAl2007} K. Park, S. Basu, W. P. King, and Z. M. Zhang, J. Quant. Spect. Rad. Transf. {\bf 109}, 305 (2008).
  \bibitem{GuoEtAl2013} Y. Guo and Z. Jacob, Opt. Express {\bf 21}, 15014 (2013).
  %hyperbolic tpv
  \bibitem{Simovski2013} C. Simovski, S. Maslovski, I. Nefedov, and S. Tretyakov, Opt. Expr. {\bf 21}, 14988 (2013)
  %natural hyperbolic materials
  \bibitem{DrachevEtAl2013} V. P. Drachev, V. A. Podolskiy, and A. V. Kildishev, Opt. Express {\bf 21}, 15048 (2013).
  \bibitem{ThompsonEtAl1998} D. W. Thompson, M. J. DeVries, T. E. Tiwald, and J. A. Woollam, Thin Solid Films {\bf 313-314}, 341 (1998).
  \bibitem{EsslingerEtAl2014} M. Esslinger, R. Vogelgesang, N. Talebi, W. Khunsin, P. Gehring, S. de Zuani, B. Gompf, and K. Kern, preprint. 
  %HMM
  \bibitem{PoddubnyEtAl2013} A. Poddubny, I. Iorsh, P. Belov, and Y. Kivshar, Nat. Photonics {\bf 7}, 948 (2013).
  \bibitem{CaiEtAl2010} W. Cai and V. Shalaev, {\em Optical Metamaterials: Fundamentals and Applications} (Springer, New York, Dordrecht, Heidelberg, London, 2010).
  %PD
  \bibitem{BasuZhang2009} S. Basu and Z. M. Zhang, Appl. Phys. Lett. {\bf 95}, 133104 (2009).
  \bibitem{BasuZhang2011} S. Basu and M. Francoeur, Appl. Phys. Lett. {\bf 99}, 143107 (2011).
  %multilayer HMM
  \bibitem{HoffmanEtAl2007} A. J. Hoffman, L. Alekseyev, S. S. Howard, K. J. Franz, D. Wasserman, V. A. Podolskiy, E. E. Narimanov, D. L. Sivco, and C. Gmachl, Nat. Mater. {\bf 6}, 946 (2007).  
  \bibitem{KrishnamoorthyEtAl2012} H. N. S. Krishnamoorthy, Z. Jacob, E. Narimanov, I. Kretzschmar, and V.M. Menon, Science {\bf 336}, 205 (2012). 
  \bibitem{LangEtAl2013} S. Lang, H. S. Lee, A. Y. Petrov, M. St\"{o}rmer, M. Ritter, and M. Eich, Appl. Phys. Lett. {\bf 103}, 21905 (2013).
  %nanowire HMM
  \bibitem{YaoEtAl2008} J. Yao, Z. Liu, Y. Liu, Y. Wang, C. Sun, G. Bartal, A. M. Stacy, and X. Zhang, Science {\bf 321}, 930 (2008).
  \bibitem{NoginovEtAl2009} M. A. Noginov, Y. A. Barnakov, G. Zhu, T. Tumkur, H. Li, and E. E. Narimanov, Appl. Phys. Lett. {\bf 94}, 151105 (2009).
  %Material Properties
  \bibitem{Adachi} S. Adachi, "III-V Compound Semiconductors" in {\em Handbook on Physical Properties of Semiconductors} (Springer, 2004), Vol. 2.
	\bibitem{OptSocAm} Optical Society of America, "Optical properties of materials, nonlinear optics, quantum optics" {\em in Handbook of Optics}, edited by M. Bass (McGraw-Hill Professional, 2009), Vol. 4.
  %HF hyperbolic nanowire materials, SM vs HM
  \bibitem{LiuShen2013} B. Liu and S. Shen, Phys. Rev. B {\bf 87}, 115403 (2013).
  \bibitem{BiehsEtAl2013} S.-A. Biehs, M. Tschikin, R. Messina, and P. Ben-Abdallah, Appl. Phys. Lett. {\bf 102}, 131106 (2013).
  %Limits of effective medium theory
  \bibitem{TschikinEtAl2013} M. Tschikin, S.-A. Biehs, R. Messina, P. Ben-Abdallah, J. Opt. {\bf 15}, 105101 (2013). 
  %Multilayer
  \bibitem{Yeh} P. Yeh, {\em Optical Waves in Layered Media}, (Wiley, Hoboken, 2005).
  %wiremedium
  \bibitem{WangbergEtAl2006} R. Wangberg, J. Elser, E. E. Narimanov, and V. A. Podolskiy, J. Opt. Soc. Am. B {\bf 23}, 498 (2006). 
  %frustrated modes
  \bibitem{Cravalho} E. G. Cravalho, C. L. Tien, and R. P. Caren, J. Heat Transfer \textbf{89}, 351 (1967).


\end{thebibliography}

\author{S. Lang}
\affiliation{Institute of Optical and Electronic Materials, Hamburg University of Technology, 21073 Hamburg, Germany}
\email{slawa.lang@tuhh.de}

\author{M. Tschikin}
\affiliation{Institut f\"{u}r Physik, Carl von Ossietzky Universit\"{a}t, D-26111 Oldenburg, Germany.}

\author{S.-A. Biehs}
\affiliation{Institut f\"{u}r Physik, Carl von Ossietzky Universit\"{a}t, D-26111 Oldenburg, Germany.}

\author{A. Yu. Petrov}
\affiliation{Institute of Optical and Electronic Materials, Hamburg University of Technology, 21073 Hamburg, Germany}

\author{M. Eich}
\affiliation{Institute of Optical and Electronic Materials, Hamburg University of Technology, 21073 Hamburg, Germany}

\title{Large penetration depth of near-field heat flux in hyperbolic media}

\keywords{surface phonon polaritons, hyperbolic metamaterials, nanoscale thermal radiation, near-field thermophotovoltaics, effective medium theory}

\begin{abstract}
We compare super-Planckian thermal radiation between phonon-polaritonic media and hyperbolic metamaterials. In particular, we determine the penetration depth of thermal photons inside the absorbing medium for three different structures: two semi-infinite phonon-polaritonic media supporting surface modes, two multilayer hyperbolic metamaterials and two nanowire hyperbolic metamaterials. We show that for hyperbolic modes the penetration depth can be orders of magnitude larger than for surface modes suggesting that hyperbolic materials are much more preferable for near-field thermophotovoltaic applications than pure phonon-polaritonic materials.
\end{abstract}
\maketitle
\date{\today}
\makeatletter
\begin{center}
\@date
\end{center}
\makeatother

%\maketitle
%\tableofcontents
%\newpage

%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%
% Introduction 
%
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%

% elevated heat flux
% surface modes vs. hyperbolic modes

Theoretically it is well-known for a long time that heat radiation at the nanoscale can surpass the blackbody limit by orders of magnitude~\cite{PvH1971,ZhangReview2009,ZhangReview2013}. Recently, this super-Planckian property has been confirmed experimentally~\cite{HuEtAl2008,ShenEtAl2009,Kralik2012}. The underlying mechanisms resulting in elevated radiative heat fluxes are nowadays well understood. In particular, surface phonon polaritons can enhance the radiative heat flux by several orders of magnitude~\cite{JoulainEtAl2005} due to the very large number of contributing modes~\cite{BiehsEtAl2010}. However, it has been shown very recently that also so called hyperbolic modes~\cite{Nefedov2011,BiehsEtAl2012,GuoEtAl2012} can lead to an enormous increase in the radiative heat flux which can be even larger than that caused by surface modes~\cite{BiehsEtAl2012}. 

% near-field thermophotovoltaics
This effect of enhanced near-field thermal radiation has several possible applications~\cite{ZhangReview2009,ZhangReview2013} as touchless cooling~\cite{Ottens2011,GuhaEtAl2012} and near-field imaging~\cite{Yannick,KittelEtAl2008,HuthEtAl2011,WorbesEtAl2013} for instance. But probably one of the most discussed is near-field thermophotovoltaics (nTPV)~\cite{MatteoEtAl2001,NarayanaswamyChen2003,ParkEtAl2007,GuoEtAl2013,LarocheEtAl06} which exploits the near-field enhancement effect for increasing the output power and the efficiency of TPV devices. Structures using hyperbolic materials for nTPV were proposed recently~\cite{Nefedov2011,Simovski2013}.

% hyperbolic materials
Surface modes are supported by many different materials, like metals and polar materials. On the other hand hyperbolic modes only exist in very few natural materials like calcite and tetradymites~\cite{DrachevEtAl2013,ThompsonEtAl1998,EsslingerEtAl2014}. But it is possible to fabricate hyperbolic metamaterials (HMMs) which support hyperbolic modes. These HMMs can be constructed by combining metals or metal-like materials with dielectrics in subwavelength structures~\cite{PoddubnyEtAl2013,CaiEtAl2010}. The main advantage of the hyperbolic modes with respect to the surface modes is that they are propagating inside the hyperbolic medium~\cite{PoddubnyEtAl2013}, whereas the surface modes are bound to the surface of the material which is one of the bottlenecks of nTPV~\cite{ParkEtAl2007}.

% aim of this letter
The purpose of this letter is to analyze the penetration depth (PD) of thermal photons~\cite{BasuZhang2009,BasuZhang2011}, resulting from the heat exchange between two bodies close to each other. This quantity is very important for possible nTPV applications. It defines the effective thickness of the layer in which electron-hole pairs can be generated. But also for cooling applications larger PDs are preferable because photons transport heat much faster than phonons. Furthermore it is better to have a larger volume where the heat is absorbed to avoid local overheating.


%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%
% Theoretical Model --- Heat Flux
%
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%

Let us consider two half spaces at slightly different temperatures $T$ and $T+ \Delta T$ with $\Delta T \ll T$ separated by a vacuum gap with the width $l$. The half spaces consist of homogeneous, isotropic or uniaxial materials with optical axes parallel to the surface normal. Then the radiative heat flux through the gap is given by $\Phi = h_0 \Delta T$ where the heat transfer coefficient (HTC) is~\cite{PvH1971}
\begin{equation}
  h_0 =  \sum_{j = \rs,\rp} \int_0^\infty \!\!\! H_0^j(\omega) \frac{\rd \omega}{2 \pi}\\
\label{Eq:htc0}
\end{equation}
introducing the spectral heat transfer coefficient (sHTC)
\begin{equation}
  H^j_0 (\omega) = \frac{\rd \Theta(\omega,T)}{\rd T} \int_0^\infty \!\!\! \mathcal{T}^j(\omega,k_\rho) k_\rho \frac{\rd k_\rho}{2 \pi}.
\label{Eq:HTC0}
\end{equation}
Here $\Theta(\omega,T) = \hbar \omega / \bigl(\exp(\hbar \omega / \kb T) - 1\bigr)$ is the Bose-Einstein function, $\omega$ is the angular frequency, $k_\rho$ is the wave vector component of the radiation parallel to the interfaces, $\kb$ is the Boltzmann constant, $\hbar$ is the reduced Planck constant and $c$ is the speed of light in vacuum. The energy transmission coefficient $\mathcal{T}^j$ for s- and p-polarized waves is given by
\begin{equation}
   \mathcal{T}^j (\omega, k_\rho; l) = \begin{cases}
                \frac{(1 - |r^j_1|^2)(1 - |r^j_2|^2)}{|1 - r^j_1 r^j_2 \exp(2 {\rm i} \gamma_0 l)|^2}, & \!\! k_\rho < k_0 \nonumber \\
                \frac{ 4 \Im(r^j_1) \Im(r^j_2) \exp(-2 |\gamma_0| l) }{ |1 - r^j_1 r^j_2 \exp(-2 |\gamma_0| l)|^2 },  & \!\! k_\rho > k_0
  \end{cases}, 
\end{equation}
where $r^\rs$ and $r^\rp$ are the Fresnel coefficients for reflection of s- and p-polarized light at the half spaces 1 and 2 and $\gamma_0 = \sqrt{k_0^2 - k_\rho^2}$ is the z component of the wavevector inside the vacuum gap with $k_0 = \omega/c$.

%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%
% Theoretical Model --- Materials
%
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%

In the following we consider the heat exchange by thermal radiation for the three different structures depicted in Fig.~\ref{Fig1:Structure}: (i) two gallium nitride (GaN) half spaces, (ii) two multilayer HMM (mHMM)~\cite{HoffmanEtAl2007,LangEtAl2013,KrishnamoorthyEtAl2012} half spaces composed of GaN/Ge bilayers and (iii) two nanowire HMM (wHMM)~\cite{YaoEtAl2008,NoginovEtAl2009} half spaces consisting of GaN nanowires immersed in a Ge host. The optical response of GaN is in the infrared dominated by the optical phonons so that the relative permittivity of GaN can be described by a Drude-Lorentz model~\cite{Adachi} $\epsilon_{\rm GaN} = \epsilon_\infty (\omega_{\rm LO}^2 - \omega^2 - \ri \omega_{\rm col} \omega)/(\omega_{\rm TO}^2 - \omega^2 - \ri \omega_{\rm col} \omega)$ with $\epsilon_\infty = 5.35$, $\omega_{\rm LO} = 141\cdot10^{12}\,{\rm rad/s}$, $\omega_{\rm TO} = 106\cdot10^{12}\,{\rm rad/s}$, and $\omega_{\rm col} = 1.52\cdot10^{12}\,{\rm rad/s}$. It is important to notice that inside the {\itshape reststrahlen band} of GaN ($\omega_{\rm TO} < \omega < \omega_{\rm LO}$) the permittivity is negative so that GaN can support surface phonon polaritons inside this frequency range. On the other hand, the permittivity of germanium (Ge) is in the infrared to very good approximation dispersionless having a value of $\epsilon_{\rm Ge} = 16$~\cite{OptSocAm}. Both materials are nonmagnetic.

\begin{figure}
  \epsfig{file = Fig1.eps, width = 0.45\textwidth}
  \caption{Systems to be analyzed. Two identical half spaces of (i) bulk GaN, of (ii) GaN/Ge layer
           HMMs and of (iii) GaN/Ge wire HMMs separated by a vacuum gap. The HMMs are modeled as
           effective media. The GaN filling factor of the layer HMM is 50 \%, the one of the wire HMM is 30 \%.
           The gap width is $l$. \label{Fig1:Structure}}
\end{figure}

%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%
% Theoretical Model --- Effective Medium Theory
%
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%

In order to describe the optical response of the structures (ii) and (iii) we use effective medium theory (EMT) which gives reliable results if the unit-cell size of the underlying structure is much smaller than the wavelength. In the far-field regime the dominant wavelength is the thermal wavelength which is about $10\,\mu{\rm m}$ at $300\,{\rm K}$. In the near-field regime the contribution of the evanescent waves sets another constraint to the applicability of the effective description. As a rule of thumb the effective description gives reliable results for distances $l$ larger than the unit-cell size~\cite{BiehsEtAl2013,TschikinEtAl2013}. For smaller distances it tends to overestimate the hyperbolic heat flux contribution~\cite{LiuShen2013,BiehsEtAl2013}. According to the EMT the effective permittivity of the mHMM is given by~\cite{Yeh}
\begin{align}
  \epsilon_\rho = \epsilon_x = \epsilon_y &= f \epsilon_{\rm GaN} + (1 - f)\epsilon_{\rm Ge}, \\
  \epsilon_z &= \biggl( \frac{f}{\epsilon_{\rm GaN}} + \frac{1 - f}{\epsilon_{\rm Ge}} \biggr)^{-1},
\end{align}
and for the wHMM it is given by~\cite{WangbergEtAl2006}
\begin{align}
  \epsilon_\rho = \epsilon_x = \epsilon_y &= \epsilon_{\rm Ge} \frac{(1 + f) \epsilon_{\rm GaN} + (1 - f)\epsilon_{\rm Ge}}{(1 - f) \epsilon_{\rm GaN} + (1 + f)\epsilon_{\rm Ge}}, \\
  \epsilon_z &= f \epsilon_{\rm GaN} + (1 - f) \epsilon_{\rm Ge},
\end{align}
where $f$ is the volume filling fraction of GaN. We choose $f$ = 50 \% for the mHMM and $f$ = 30 \% for the wHMM. z is the direction perpendicular to the surfaces. Note that although we call the structures (ii) and (iii) HMMs they are hyperbolic only in certain frequency ranges. For the mHMM the range is $106-141\cdot10^{12}\,{\rm rad/s}$ and for the wHMM it is $106-121\cdot10^{12}\,{\rm rad/s}$ (with a small non-hyperbolic region from $111-112\cdot10^{12}\,{\rm rad/s}$). In this paper only effective media are considered. Real structures are to be studied in later publications.

%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%
% Theoretical Model --- Penetration Depth 
%
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%

Now, we want to define the PD $\delta$ of the thermal radiation into the colder medium. To this end, we need to determine the heat flux or the heat transfer coefficient inside the colder medium. It is already clear that due to the rotational symmetry of the problem the heat flux is along the z direction only. Furthermore, the components of the electric and magnetic fields parallel to the interfaces are continuous which implies that the z component of the Poynting vector is continuous at the interface as well. Finally, we know that the intensity of a plane wave with a given frequency $\omega$ and tangential wavevector $k_\rho$ impinging on a semi-infinite uniaxial medium (with the optical axis along the surface normal) is damped by a factor $\exp(- 2 \Im(\gamma^j) z)$ where $\gamma^j$ is the polarization dependent wavevector component along z direction, i.e. the direction of propagation of the heat flux. Hence, from Eq.~(\ref{Eq:HTC0}) giving the spectral heat transfer coefficient inside the vacuum gap and in particular at the interface of the absorbing medium, we can easily deduce the sHTC inside the colder medium. We obtain
\begin{equation}
       H^j (\omega; z) = \frac{\rd \Theta(\omega,T)}{\rd T} \int_0^\infty \!\!\! \mathcal{T}^j(\omega,k_\rho) \re^{-2 \Im(\gamma^j) z} k_\rho \frac{\rd k_\rho}{2 \pi}.
\end{equation}
assuming the vacuum / cold medium interface is located at z = 0. By replacing $H^j_0$ by $H^j$ in Eq.~(\ref{Eq:htc0}) we obtain the heat transfer coefficient $h(z)$ inside the colder medium. The z component of the wavevector is in our case given by
\begin{equation}
  \gamma^\rs = \sqrt{k_0^2 \epsilon_\rho- k_\rho^2 }
\end{equation}
for s-polarized waves ({\itshape ordinary waves}) and by
\begin{equation}
  \gamma^\rp = \sqrt{k_0^2 \epsilon_\rho - k_\rho^2 \frac{\epsilon_\rho}{\epsilon_z} }
	\label{Eq:gammaP}
\end{equation}
for p-polarized waves ({\itshape extra-ordinary waves}). Obviously, the anisotropy makes itself felt through the p-polarized waves. For $\epsilon_\rho \epsilon_z < 0$  the p-polarized waves have hyperboloidal isofrequency curves in k-space~\cite{PoddubnyEtAl2013}. The above introduced mHMM and wHMM support broad frequency bands for such hyperbolic modes. 

The PD $\delta$ is now defined such that it determines the distance at which the heat transfer coefficient $h$ has dropped to $1/\re$ of its value inside the vacuum gap, i.e.\ $h(\delta) = h(0)/\re = h_0/\re$. In a similar way we define the spectral PD $\delta(\omega)$ as the distance at which the spectral heat transfer coefficient $H = H^\rs + H^\rp$ drops to $1/\re$, i.e.\ $H(\omega,\delta(\omega)) = H(\omega,0)/\re = H_0(\omega)/\re  = (H^\rs_0(\omega)+H^\rp_0(\omega))/\re$.

%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%
% Discussion --- Spectral Heat Flux 
%
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\begin{figure*}[Hhbt]
  \epsfig{file = Fig2.eps, width = 0.9\textwidth}
  \caption{Spectral thermal PD (solid) and sHTC in the vacuum gap (dashed) for (a) bulk GaN, (b) GaN/Ge layer HMMs and (c) GaN/Ge wire HMMs. The sHTC is normalized to the sHTC between black bodies. The gap width is $l = 10\,{\rm nm}$.
           \label{Fig2:sHTC}}
\end{figure*}


Before discussing the total PD for the different systems (i)-(iii) we present in Fig.~\ref{Fig2:sHTC} the sHTC $H_0(\omega)$ and the spectral PD $\delta(\omega)$ for the different structures choosing $l = 10\,{\rm nm}$ and $T = 300\,{\rm K}$. It can be seen in Fig.~\ref{Fig2:sHTC}(a) that the heat flux between bulk GaN has a large, narrow peak at the surface mode frequencies as expected in this case~\cite{JoulainEtAl2005}. For the two HMMs in Fig.~\ref{Fig2:sHTC}(b) and (c) the heat flux is strong in the broad frequency bands where hyperbolic modes exist. These bands are due to the so called type I and type II responses~\cite{PoddubnyEtAl2013}. For the mHMM in Fig.~\ref{Fig2:sHTC}(b) both bands touch each other which is due to the choice of $f = 0.5$ resulting in a broad plateau for the sHTC and the spectral PD. For the wHMM in Fig.~\ref{Fig2:sHTC}(c) both bands are separated by a small, non-hyperbolic region (when choosing $f = 1/3$ instead of $f = 0.3$ both bands would touch as well). That is the reason for the small dip in the middle  of the plateau of the sHTC.

For all structures the spectral PD decreases when the sHTC increases. This is not surprising if Eq.~(\ref{Eq:gammaP}) is recalled. The larger $k_\rho$ the larger is the imaginary part of $\gamma^\rp \approx i k_\rho \sqrt{\epsilon_\rho / \epsilon_z}$ ($k_\rho \gg k_0$). And the large heat fluxes are caused by p-polarized, high $k_\rho$ modes. For the HMMs the imaginary part of $\gamma^\rp$ is due to the imaginary parts of $\epsilon_\rho$ and $\epsilon_z$ because in a lossless case this term $\sqrt{\epsilon_\rho / \epsilon_z}$ is just an imaginary number and the modes are not damped at all. But for GaN the term is 1 and damping is always observed independent of losses. Comparing the results in Fig.~\ref{Fig2:sHTC} for the different structures (i)-(iii) it becomes apparent that the PD in the HMMs can have values as small as the PD in GaN at the peak frequency of the sHTC. However, considering the whole hyperbolic region $\delta(\omega)$ can for the HMMs also have values which are one or even three orders of magnitude larger than the corresponding value for bulk GaN. By optimizing the HMMs to have smaller absorption this difference in PD can be made even larger.

%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%
% Discussion --- Total Heat Flux 
%
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%

\begin{figure}[Hhbt]
  \epsfig{file = Fig3.eps, width = 0.45\textwidth}
  \caption{HTC in the vacuum gap (a) and thermal PD (b) at $T = 300\,{\rm K}$ for the systems: bulk GaN, GaN/Ge layer HMM, GaN/Ge wire HMM and bulk Ge. The HTC is normalized to the HTC between black bodies.
           \label{Fig3:PDandHTC}}
\end{figure}

In Fig.~\ref{Fig3:PDandHTC}(a) the total HTC for the three systems and additionally between two bulk Ge half spaces is shown. It is normalized to the HTC between two black bodies $h_{\rm BB}$ which is about $6.12\,{\rm W}/({\rm m}^2 {\rm K})$ independent of the gap width. The HMMs and GaN show the well known $1/l^2$ dependence in the near field. It should be emphasized that real HMMs have a cutoff in the k-space which depends on the unit-cell size of the structure resulting in a saturation of the heat flux for distances $l$ smaller than the unit-cell size~\cite{BiehsEtAl2013}. The relative heat flux of Ge approaches the value $\epsilon_{\rm Ge} = 16$ because the heat transfer is sustained by frustrated total internal reflection modes only which contribute by photon tunneling~\cite{Cravalho}. Since we have neglected losses for Ge its PD is infinite. Of course, real Ge has losses due to imperfections but the PD $\delta$ can still be very large. As can be observed in Fig.~\ref{Fig3:PDandHTC}(a) the wHMM has a relatively weak heat transfer coefficient compared to bulk GaN and the mHMM. However, we have checked that by replacing Ge with some dielectric having a lower refraction index one can make the hyperbolic band much broader and thus enhance the near-field heat flux significantly~\cite{BiehsEtAl2012}.

Finally, the total PD $\delta$ is depicted in Fig.~\ref{Fig3:PDandHTC}(b). In the far-field and the intermediate region it is more or less constant for all materials. The higher the filling factor of GaN the more lossy is the effective material and the shorter the penetration. To get an approximate boundary between far and near field we calculated the distance $l$ at which 50 \% of the total heat flux is due to hyperbolic modes. This distance is 115 nm for the mHMM and 44 nm for the wHMM --- so approximately given by the start of the $1/l^2$ dependence of the HTC. The spectral heat flux is generally much broader (e.g. for two black body half spaces) than the hyperbolic bands such that they only dominate in the near field on which our focus lies in this paper. In this strong near-field regime where the surface or hyperbolic modes dominate the heat flux for GaN and the HMMs, $\delta$ drops dramatically. This behavior can be understood by the fact that with smaller $l$ modes with $k_\rho$ on the order of $1/l$ dominate the thermal radiation so that $\Im(\gamma^j) \propto k_\rho \propto 1/l$ and hence the PD becomes approximately proportional to $l$. Most importantly in the intermediate regime the PD in the HMMs can be two to three orders of magnitude larger than in GaN and in the strong near-field regime it can be more than one order of magnitude larger than in GaN. Hence, our numerical results suggest that hyperbolic materials are preferable to phonon-polaritonic media when larger near-field PDs are needed as in the case of nTPV~\cite{ParkEtAl2007}.

It should be mentioned that the structures presented here are not directly applicable to nTPV systems. In this paper we report on the general property of hyperbolic modes supporting large penetration depths of near-field heat flux. For nTPV applications the materials should produce hyperbolic regions in the near infrared where photovoltaic cells are available. Concrete systems are left for further studies.


%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%
% Conclusion
%
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%

In conclusion, we have shown with concrete examples that the PD of thermal photons in the near-field regime can be much larger for materials or systems supporting hyperbolic modes than for materials supporting surface modes. The reason is the different nature of those modes. The penetration depth is potentially an important property for a variety of applications including touchless cooling and thermophotovoltaics.

%\acknowledgement
The authors from Hamburg University of Technology gratefully acknowledge financial support from the German Research Foundation (DFG) via SFB 986 $"M^3"$, project C1. M. T. gratefully acknowledges support from the Stiftung der Metallindustrie im Nord-Westen. S.-A. B. and M. T. acknowledge financial support by the DAAD and Partenariat Hubert Curien Procope Program (project 55923991).



\end{document}
